Author's response to reviews

Title: Irregular breakfast eating and health status among adolescents in Taiwan

Authors:

Rea-Jeng Yang (rea@ntcn.edu.tw)
Edward K Wang (mymcwang@yahoo.com.tw)
Yeu-Sheng Hsieh (ysh@ntu.edu.tw)
Mei-Yen Chen (meiyen@mail.cgit.edu.tw)

Version: 5 Date: 25 November 2006

Author's response to reviews: see over
Manuscript number: **MS: 7508125591012280**

To editor,

Thank you for your suggestions. We’ve followed your comments and made some revision to fit the appropriate format.

1. We’ve removed the text 'Title:' from the title page.

2. We’ve removed the authors' qualifications and job titles from the title page.

3. We’ve removed the repeated title above the Abstract.

4. We’ve removed the keywords section from the manuscript.

5. Regarding the reference list. We have changed the format to fit the journal requirement, e.g. contain all author’s names.

6. We have changed all colored text to black.

7. We have stated that all authors have read and approved the final manuscript.

8. We have stated ethical approval for the study, which included the ethics committee's full name on p3.